Author's response to reviews

Title: Atypical presentation of hepatocellular carcinoma: a mass on the left thoracic wall

Authors:

Sahin Coban (scoban72@yahoo.com)
Osman Yuksel (ottomanyuksel@hotmail.com)
Seyfettin Koklu (gskoklu@yahoo.com)
Koray Ceyhan (korayceyhan@yahoo.com)
Meltem Baykara (meltembaykara@yahoo.com)
Abdulkadir Dokmeci (akdokmeci@hotmail.com)

Version: 5 Date: 7 November 2004

Author's response to reviews:

Dear Sir;

We tried to revise the manuscript according to your suggestions. The manuscript was edited by someone whose language is native English and who is expert on medical manuscripts. Though you have further suggestions please mark these writing errors.

With kind regards.

Sahin Coban